\documentclass[11pt,twoside,reqno]{amsart}
\usepackage{amsmath, amsthm, amscd, amsfonts, amssymb, graphicx, color}
\usepackage[bookmarksnumbered, plainpages]{hyperref}
%=============================
\usepackage{lipsum,amssymb,graphicx,youngtab,ytableau,tikz,subcaption}
\hypersetup{  colorlinks, linkcolor=blue, urlcolor=blue, citecolor=green}
%====================================================

%===================================
\usepackage{fancyhdr}
\usepackage{lastpage}
\usepackage{caption}
\usepackage{subcaption}
\usepackage{cite}
\usepackage[footskip=0.25in]{geometry}

\usepackage{etoolbox}
\makeatletter
\def\@settitle{\begin{center}%
		\baselineskip14\p@\relax
		\bfseries
		\normalfont\LARGE%<- NEW
		%\uppercasenonmath\@title
		\@title
	\end{center}%
}
%\patchcmd{\@settitle}{\uppercasenonmath\@title}{}{}{}
\patchcmd{\@setauthors}{\MakeUppercase}{}{}{}
%\patchcmd{\section}{\scshape}{}{}{}
\def\section{\@startsection{section}{1}%
    \z@{.7\linespacing\@plus\linespacing}{.5\linespacing}%
    %  {\normalfont\scshape\centering}}% DELETED
    {\normalfont\bf}}% NEW
\def\subsection{\@startsection{subsection}{1}%
    \z@{\linespacing\@plus\linespacing}{.5\linespacing}%
    %  {\normalfont\centering}}% DELETED
    {\normalfont\it}}% NEW
\def\subsubsection{\@startsection{subsubsection}{1}%
    \z@{\linespacing\@plus\linespacing}{.5\linespacing}%
    %  {\normalfont\centering}}% DELETED
    {\normalfont\it}}% NEW
\makeatother

\addtolength{\topmargin}{-1.5cm} \linespread { 1.3} \textwidth15.7
cm \textheight 23 cm \addtolength{\hoffset}{-0.3cm} \oddsidemargin
0cm \evensidemargin 0cm \setcounter{page}{1}

%------------------------------------------------------------
%%%%%%%%%%%

%----------------------%
\newtheorem{theorem}{Theorem}[section]
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{notation}{Notation}
\newtheorem{conjecture}[theorem]{Conjecture}
\theoremstyle{definition}
\newtheorem{remark}[theorem]{Remark}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{example}[theorem]{Example}


\newcommand{\centered}[1]{\begin{tabular}{l} #1 \end{tabular}}

%
%\renewcommand{\footrulewidth}{0.05pt}
%\pagestyle{fancyplain}
%\fancyhf{}
%\lfoot{ \fancyplain{}{A. B. Khalaf and P. Padma, On Certain Types of Neutrosophic Fuzzy Graphs } }

\renewcommand{\footrulewidth}{0.05pt}
\pagestyle{fancyplain} 
\fancyhf{} 
\lhead{
%\fancyplain{}
{A. B. Khalaf and P. Padma, On Certain Types of Neutrosophic Fuzzy Graphs} } \rhead{ \fancyplain{}{\thepage\ of
\pageref{LastPage}} } \lfoot{ \fancyplain{}{} }

\renewcommand{\qedsymbol}{$\blacksquare$}

\makeatletter
%\patchcmd{\@settitle}{center}{flushleft}{}{}
%\patchcmd{\@settitle}{center}{flushleft}{}{}
\patchcmd{\@setauthors}{\centering}{\raggedright}{}{}
\patchcmd{\abstract}{3pc}{0pt}{}{} % remove indentation
\makeatother
%%------------------------------------------------------------------------------------%
\begin{document}
\begin{thebibliography}{10}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\bibitem{akm} 
Akram M, Akmal R (2016) Operations on intuitionistic fuzzy graph
structures. Fuzzy Inf Eng 8(4):389–410
\bibitem{ak}
Akram M.; Davvaz B. Strong intuitionistic fuzzy graphs. In: Filomat, vol. 26, no. 1 (2012) 177-–196.
\bibitem{ak1}
 Akram M; and  Dudek W. A. Interval-valued fuzzy graphs. In: Computers and Mathematics with Applications, vol. 61, no. 2 (2011) 289-–299.
\bibitem{ak2}
Akram M. Interval-valued fuzzy line graphs. In: Neural Comp. and Applications, vol. 21 (2012) 145–-150.
\bibitem{ak3}
Akram M. Bipolar fuzzy graphs. In: Information Sciences, vol. 181, no. 24 (2011) 5548–-5564.
\bibitem{ak4} 
Akram M. Bipolar fuzzy graphs with applications. In: Knowledge Based Systems, vol. 39 (2013) 1-–8.
\bibitem{ak5}
Akram M.; Adeel A. m-polar fuzzy graphs and m-polar fuzzy line graphs. In: Journal of Discrete Mathematical Sciences and Cryptography, 2015.
\bibitem{ak6}
Akram M.;  Dudek W. A. Regular bipolar fuzzy graphs. In: Neural Computing and Applications, vol. 21(2012), 97--205.
\bibitem{ak7}
Akram M.; Dudek W. A.;  Sarwar S. Properties of Bipolar Fuzzy Hypergraphs. In: Italian Journal of Pure and Applied Mathematics no. 31 (2013), 141--160.
\bibitem{ak8}
Akram M.; Alshehri N. O.; Dudek W. A. Certain Types of Interval-Valued Fuzzy Graphs. In : Journal of Appl. Mathematics, 2013, 11 pages, {\url{http://dx.doi.org/10.1155/ 2013/857070}}.
\bibitem{ak9}
Akram M.; Yousaf M. M. ; Dudek W. A. Self- centered interval-valued fuzzy graphs. In: Afrika Matematika, vol. 26, Issue 5(2015), 887--898.
\bibitem{ak10}
Akram M, Siddique S (2017) Neutrosophic competition graphs
with applications. J Intell Fuzzy Syst 33(2):921–935
\bibitem{ak11}
Akram M, Waseem N, Dudek WA (2017) Certain types of edge
m-polar fuzzy graphs. Iran J Fuzzy Syst 14(4):27–50
%==========================================================
\bibitem{ata}
Atanassov K. Intuitionistic fuzzy sets. In: Fuzzy Sets and Systems. vol. 20 (1986) 87-96.
\bibitem{ata1}
Atanassov K.; Gargov G. Interval valued intuitionistic fuzzy sets. In: Fuzzy Sets and Systems, vol. 31 (1989) 343--349.
\bibitem{ba}
 Bhattacharya P. Some remarks on fuzzy graphs. In: Pattern Recognition Letters 6 (1987) 297--302.
\bibitem{bor}
Borzooei R. A.; Rashmanlou H.  More Results On Vague Graphs. U.P.B. Sci. Bull., Series A, Vol. 78, Issue 1 (2016), 109--122.
\bibitem{br1}
Broumi S.; Smarandache F.; Talea M.; and Bakali A. Single Valued Neutrosophic Graphs: Degree, Order and Size, In 2016 IEEE international conference on fuzzy systems (FUZZ-IEEE), IEEE 2016, 2444--2451. \href{https://scholar.google.com/scholar?hl=en&as_sdt=0%2C5&q=Single+Valued+Neutrosophic+Graphs%3A+Degree%2C+Order+and+Size+&btnG=}{Google Scholar} 
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\bibitem{bro}
Broumi S.; Talea M.; Bakali A; Smarandache F. Single Valued Neutrosophic Graphs. In: Journal of New Theory, no. 10, 68-101 (2016).
\bibitem{br2}
Broumi S.; Smarandache F.; Bakali A; Mehra S.; Talea M.; Singh M. Strong Degrees in Single Valued Neutrosophic Graphs,  Future of Information and Communication Conference, Springer, Cham, 2018, 221--238. \href{https://scholar.google.com/scholar?hl=en&as_sdt=0%2C5&q=Strong+Degrees+in+Single+Valued+Neutrosophic+Graphs&btnG=}{Google Scholar}
%=============================================================
 \bibitem{gh}
 Ghorai G, Pal M (2017) Certain types of product bipolar fuzzy
graphs. Int J Appl Comput Math 3(2):605–619
%=============================================================
\bibitem{br3}
Broumi S.; Talea M.; Bakali A; Smarandache F. Interval Valued Neutrosophic Graphs. SISOM Conference (2016), in press.
\bibitem{br4}
Broumi S.; Smarandache F.; Talea M.; and Bakali A.  An Introduction to Bipolar Single Valued Neutrosophic  Graph Theory. OPTIROB conference, 2016.
\bibitem{br5}
Broumi S.; Talea M.; Bakali A; Smarandache F. Operations on Interval Valued Neutrosophic Graphs. New Trends in Neutrosophic Theory and Applications (2016), 231--254. \href{https://scholar.google.com/scholar?hl=en&as_sdt=0%2C5&q=Operations+on+Interval+Valued+Neutrosophic+Graphs&btnG=}{Google Scholar}
\bibitem{br6}
Broumi S.; Talea M.; Bakali A; Smarandache F. On Strong Interval Valued Neutrosophic Graphs. Critical Review. Volume XII (2016), 49--71. \href{https://scholar.google.com/scholar?hl=en&as_sdt=0%2C5&q=On+Strong+Interval+Valued+Neutrosophic+Graphs&btnG=}{Google Scholar}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\bibitem{dev}
Devadoss A. V.; Rajkumar A.; Praveena N. J. P. A Study on Miracles through Holy Bible using Neutrosophic Cognitive Maps (NCMS). In: International Journal of Computer Applications, 69(3) (2013).
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\bibitem{di}
Divya G.;  Malarvizhi J. Some operations on neutrosophic fuzzy graphs, International Journal of Mathematical Archive  8(9)(2017), 120--125.   
\href{https://scholar.google.com/scholar?hl=en&as_sdt=0%2C5&q=SOME+OPERATIONS+ON+NEUTROSOPHIC+FUZZY+GRAPHS&btnG=}{Google Scholar} 
%============================================================
\bibitem{mis}
Mishra S. N.; Pal A. Product of Interval Valued Intuitionistic fuzzy graph. In: Annals of Pure and Applied Mathematics, Vol. 5, No.1 (2013), 37--46.
%====================================================================================================
\bibitem{na}
 Nagoor Gani A.; Ahamed M.B. Order and Size in Fuzzy Graphs. Bulletin of Pure and Applied Sciences 22(1)(2003), 145--148.\href{https://scholar.google.com/scholar?hl=en&as_sdt=0%2C5&q=Order+and+Size+in+Fuzzy+Graphs&btnG=}{Google Scholar} 

\bibitem{nag1}
Nagoor Gani A.; Shajitha Begum S. Degree, Order and Size in
Intuitionistic Fuzzy Graphs. In: Intl. Journal of Algorithms, Computing and Mathematics, (3)3 (2010).
\bibitem{nag2}
Nagoor Gani A.; Latha S. R. On Irregular Fuzzy Graphs. In: Applied Mathematical Sciences, Vol. 6, no. 11 (2012) 517--523.
\bibitem{nag3}
Nagoor Gani A.;  Malarvizhi J. Isomorphism Properties on Strong Fuzzy Graphs. International Journal of Algorithms, Computing and Mathematics 2009, 39--47.
 \bibitem{naz}
 Naz S, Rashmanlou H, Malik MA (2017) Operations on single
valued neutrosophic graphs with application. J Intell Fuzzy Syst
32(3):2137–2151

%===================================================================================
\bibitem{pa}
Parvathi R.; and  Karunambigai M. G. Intuitionistic Fuzzy Graphs. In: Computational Intelligence. In : Theory and applications, International Conference in Germany, Sept 18 -20, 2006.
\bibitem{pa1}
Parvathi R, Karunambigai MG (2006) Intuitionistic fuzzy graphs.
In: Computational intelligence, theory and applications, Springer,
New York, pp 139–150.
\bibitem{pa2}
Parvathi R, Karunambigai MG, Atanassov Krassimir T (2009)
Operations on intuitionistic fuzzy graphs. In: 2009 IEEE interna-
tional conference on fuzzy systems, pp 1396–1401
\bibitem{pa3}
Parvathi R, Thamizhendhi G (2010) Domination in intuitionistic
fuzzy graphs. Notes Intuit Fuzzy Sets 16(2):39–49
\bibitem{ra}
Radhamani C.; Radhika C. Isomorphism on Fuzzy Hypergraphs, IOSR Journal of Mathematics (IOSRJM), ISSN: 2278-5728 Volume 2, Issue 6 (Sep.-Oct. 2012), 24--31.


\bibitem{ra2}Rashmanlou H, Borzooei RA, Samanta S, Pal M (2016) Properties
of interval valued intuitionistic (s, t)–fuzzy graphs. Pac Sci Rev A
Nat Sci Eng 18(1):30–37
16. 
\bibitem{ra3}
Rashmanlou H, Samanta S, Pal M, Borzooei RA (2015) Bipolar
fuzzy graphs with categorical properties. Int J Comput Intell Syst
8(5):808–818
17. 
\bibitem{ra4}
Rashmanlou H, Samanta S, Pal M, Borzooei RA (2015) Intu-
itionistic fuzzy graphs with categorical properties. Fuzzy Inf Eng
7(3):317–334
\bibitem{sa}
Sahoo S.; Pal M.  Different types of products on intuitionistic
fuzzy graphs. Pac. Sci. Rev. A Nat. Sci. Eng. 17(3), (2015)87–96
\bibitem{sh}
Shannon A, Atanassov K.  On a generalization of intuition-
istic fuzzy graphs. NIFS 12(1), (2006)24–29.

\bibitem{sm}
Smarandache F. Types of Neutrosophic Graphs and neutrosophic
Algebraic Structures together with their Applications in Technology, seminar. Universitatea Transilvania din Brasov, Facultatea de Design de Produs si Mediu, Brasov, Romania 06 June 2015. \href{https://scholar.google.com/scholar?cluster=10233498251875564189&hl=en&oi=scholarr}{Google Scholar} 
\bibitem{sma}
Smarandache F (2020) Extension of hypergraph to n-
superhypergraph and to plithogenic n-superhypergraph, and
extension of hyperalgebra to n-ary (classical-/neutro-/anti-)
hyperalgebra. Neutrosophic Sets Syst 33(1):290–296
\bibitem{sma1}
Smarandache F. Refined Literal Indeterminacy and the Multiplication Law of
Sub-Indeterminacies. In: Neutrosophic Sets and Systems, Vol. 9 (2015) 58--63.
%%%%%%%%%%%%%%%%%%%%%%%%%%

\bibitem{sma2}
Smarandache F. Symbolic Neutrosophic Theory. Brussels: Europanova, 2015,195 p.
\bibitem{sma3}
Smarandache F. Neutrosophic set- a generalization of the intuitionistic fuzzy set. In: Granular Computing, 2006 IEEE Intl. Conference, (2006) 38 -- 42, {\url{DOI: 10.1109/GRC. 2006.1635754}}.
\bibitem{tur}
Turksen I. B. Interval valued fuzzy sets based on normal forms. In: Fuzzy Sets and Systems, vol. 20(1986), 191--210. \href{https://scholar.google.com/scholar?hl=en&as_sdt=0%2C5&q=Interval+valued+fuzzy+sets+based+on+normal+forms.&btnG=}{Google Scholar}
\bibitem{wa}
Wang H.; Smarandache F.; Zhang Y.; Sunderraman R. Single Valued  Neutrosophic  Sets. Multispace and Multistructure 4 (2010), 410-413. \href{https://scholar.google.com/scholar?hl=en&as_sdt=0%2C5&q=Single+valued++Neutrosophic++Sets+by+Wang&btnG=}{Google Scholar}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\bibitem{wa1}
Wang H.; Smarandache F.; Zhang Y.; Sunderraman R. Interval Neutrosophic Sets and Logic: Theory and Applications in Computing. Phoenix: Hexis, 2005.


%=============================================================
\bibitem{ya}
Yang H-L, Guo Z-L, She Y, Liao X (2016) On single valued neu-
trosophic relations. J Intell Fuzzy Syst 30(2):1045–1056
\bibitem{ye}
Ye J (2014) Single-valued neutrosophic minimum spanning tree
and its clustering method. J Intell Syst 23(3):311–324
%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%=============================================================
\bibitem{za}
Zadeh L. Fuzzy sets. In: Information and Control, 8 (1965) 338--353.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\bibitem{ze}
Zeng S.; Shoaib M.; Ali S.; Smarandache F.; Rashmanlou H.;  Mofidnakhaei F. Certain Properties of Single-Valued Neutrosophic Graph With Application in Food and Agriculture Organization, International Journal of Computational Intelligence Systems (2021). \href{https://scholar.google.com/scholar?hl=en&as_sdt=0%2C5&q=Certain+Properties+of+Single-Valued+Neutrosophic+Graph+With+Application+in+Food+and+Agriculture+Organization&btnG=}{Google Scholar}

\end{thebibliography}



\tolerance=10000 \baselineskip18truept
\newbox\thebox
\global\setbox\thebox=\vbox to 0.2truecm{\hsize
0.15truecm\noindent\hfill}
\def\boxit#1{\vbox{\hrule\hbox{\vrule\kern0pt
     \vbox{\kern0pt#1\kern0pt}\kern0pt\vrule}\hrule}}
\def\qed{\lower0.1cm\hbox{\noindent \boxit{\copy\thebox}}\bigskip}
\def\ss{\smallskip}
\def\ms{\medskip}
\def\nt{\noindent}
\hspace{1cm}
%\begin{tabular}{lcr}
%    \includegraphics[width=0.1\linewidth]{NSS}&\hspace{7cm} &  \scriptsize Neutrosophic Sets and Systems,  Vol. {\bf\rm xx}, {\bf\rm 20xx}\\ \hline \vspace*{-5mm}\\
 %   \includegraphics[width=0.1\linewidth]{NMU}&\hspace{7cm} & \begin{tabular}{r} \scriptsize University of New
 %   Mexico \\
%     \includegraphics[width=0.1\linewidth]{NSS1}\end{tabular}
%\end{tabular}
\renewcommand\footnoterule{\rule{\linewidth}{1pt}}
\let\thefootnote\relax



\title{ On Certain Types of Neutrosophic Fuzzy Graphs}
\author{Alias B. Khalaf$^{1,*}$ and   prithivirajan Padma $^2$ \\
    \footnotesize{$^{1}$Department of Mathematics, College of Science,
University of Duhok, Iraq.; {aliasbkhalaf@uod.ac}}\\
    \footnotesize{$^{2}$Department of Mathematics, Cheran Arts Science College, Kangeyam ; {padmaprithivirajan@gmail.com}}\\
    %\footnotesize{$^{3}$Affiliation 3; e-mail@e-mail.com}\\
    \footnotesize{$^{*}$Correspondence: \url{aliasbkhalaf@uod.ac}(Alias B. Khalaf)}
    \footnote{A. B. Khalaf and P. Padma, On Certain Types of Neutrosophic Fuzzy Graphs}
}


\maketitle




\renewcommand\abstractname{\textbf{Abstract}}
\begin{abstract}
In this paper, we introduce some types of NF graphs and operations. Also we define the partial NF subgraph, spanning NF subgraph, strong degree of the vertex, total strong degree of the vertex and its properties are included.
\medskip     \\
{\bf  Keywords:} neutrosophic fuzzy graphs; neutrosophic fuzzy degree; neutrosophic fuzzy total degree; strong edges; strong degree, neutrosophic sum.  \\
\noindent {\sf 2020 Mathematics Subject Classification:} 03E72, 05C72, 05C78, 05C99.\\
{\bf \hrule 
}
\end{abstract}
\thispagestyle{empty}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\section{Introduction}
The notion of neutrosophic sets (NSs) was proposed by Smarandache \cite{sma3} as a generalization of  the fuzzy sets \cite{za}, intuitionistic fuzzy sets \cite{ata}, interval valued fuzzy set \cite{tur} and interval-valued intuitionistic fuzzy sets \cite{ata1} theories. The neutrosophic set is a powerful mathematical tool for dealing with incomplete, indeterminate and inconsistent  information in real world. The neutrosophic sets are characterized by a truth-membership function (t), an indeterminacy-membership function (i) and a falsity- membership function (f) independently, which are within the real standard or nonstandard unit interval $]^{-}0,1^{+} [$. In order to conveniently use NS in real life applications, Wang et al. \cite{wa} introduced the concept of the  single-valued neutrosophic set (SVNS), as a subclass of the neutrosophic sets. The same authors \cite{wa1} introduced the concept of the interval valued neutrosophic set (IVNS), which is more precise and flexible than the single valued neutrosophic set. The IVNS is a generalization of the single valued neutrosophic set, in which the three membership functions are independent and their value belong to the unit interval $[0, 1]$.
Graph theory has now become a major branch of applied mathematics and it is generally regarded as a branch of combinatorics. Graph is a widely used tool for solving combinatorial problems in different areas such as geometry, algebra, number theory, topology, optimization and computer science. If one has uncertainty regarding either the set of vertices or edges, or both, the model becomes a fuzzy graph. The extension of fuzzy graph \cite{na, nag2, ba} theory have been developed by several researchers, for instance  vague graphs \cite{bor}, considering the vertex sets and edge sets as vague sets; intuitionistic fuzzy graphs \cite{nag1, ak, pa}, considering the vertex sets and edge sets as intuitionistic fuzzy sets; interval valued fuzzy graphs \cite{ak1,ak2, ak8,ak9}, considering the vertex sets and edge sets as interval valued fuzzy sets; interval valued intuitionistic fuzzy graphs \cite{di}, considering the vertex sets and edge sets as interval valued intuitionistic fuzzy sets; bipolar fuzzy graphs \cite{ak3, ak4, ak6, ak7}, considering the vertex sets and edge sets as bipolar fuzzy sets; m-polar fuzzy graphs \cite{ak5}, considering the vertex sets and edge sets as m-polar fuzzy sets. But, if the relations between nodes (or vertices) in problems are indeterminate, the fuzzy graphs and their extensions fail. For this purpose, Smarandache \cite{sm, sma, sma2, dev}, defined four main categories of neutrosophic graphs; two are based on literal indeterminacy (I), called: I-edge
neutrosophic graph and I-vertex neutrosophic graph, deeply studied and gaining popularity among the researchers due to their applications via real world problems \cite{dev, mis}, the two others are based on $(t, i, f)$ components, called: $(t, i, f)$-edge neutrosophic graph and $(t, i, f)$-vertex neutrosophic graph, concepts not developed at all by now.
Broumi et al. \cite{bro} introduced a third neutrosophic graph model, which allows the attachment of truth-membership (t), indeterminacy-membership (i) and falsity-membership degrees (f) both to vertices and edges, and investigated some of their properties. The third neutrosophic graph model is called the single valued neutrosophic graph (SVNG for short). The single valued neutrosophic graph is a generalization of fuzzy graph and intuitionistic fuzzy graph. Also, the same authors \cite{br1} introduced neighborhood degree of a vertex and closed neighborhood degree of a vertex in single valued neutrosophic graph as a generalization of neighborhood degree of a vertex and closed neighborhood degree of a vertex in fuzzy graph and intuitionistic fuzzy graph. Recently, Broumi et al. \cite{br3, br5, br6} introduced the concept of interval valued neutrosophic graph as a generalization of fuzzy graph, intuitionistic fuzzy graph and single valued neutrosophic graph and discussed some of their properties with proof and examples. Shannon and Atanassov \cite{sh} presented the concept of relationship between IFS. Then, they have introduced the concept of intuitionistic fuzzy graphs and presented many theorems in \cite{sh}.  Parvathi et al. \cite{pa1,pa2,pa3} proposed some operations between two intuitionistic
fuzzy graphs. In \cite{ra} Rashmanlou et al. proposed many
products operations such as lexicographic, direct product,
strong product, and semi-strong product on intuitionistic
fuzzy graphs. They have described the cartesian production,
join, composition and union on intuitionistic fuzzy graphs in
their paper. For further study on intuitionistic fuzzy graphs,
please refer to \cite{ak2, ra, ra2, sa}. Smarandache \cite{sm} has 
introduced the n-SuperHyperGraph, with super-vertices that is
the most general form of graph as today. Akram et al. \cite{ak3, ak4, ak5}
have introduced the idea of pythagorean fuzzy graph.
They have described the several applications of pythagorean
fuzzy graph in their paper. Neutrosophic graph \cite{gh} is used to
model many real-world problem which consists of inconsistent information. Recently, many scientists have researched on graph in neutrosophic environment, for instance, Yang et al. \cite{ye}, Akram \cite{ak10, ak11}, Ye \cite{ye}, Naz et al. \cite{na} and Broumi \cite{br1}. Throughout the paper we denote Neutrosophic fuzzy graph by NF-graph. 
In this paper, we introduce some types of NF graphs and operations. Also we define the partial NF subgraph, spanning NF subgraph, strong degree of the vertex, total strong degree of the vertex, neutrosophic sum and its properties are included.


%========================
\section{Preliminaries}
In this section, we mainly recall some notions related to neutrosophic sets, single valued neutrosophic sets, single valued neutrosophic graphs, relevant to the present article.
See \cite{sma3, wa} for further details and background.
\begin{definition}\cite{sma3}
Let $X$ be a space of points (objects) with generic elements in $X$ denoted by $x$; then, the neutrosophic set $A$ (in short NS $A$) is an object having the form   $A= \{[x : T_A(x), I_A(x), F_A(x)], x \in X\} $ where the functions  $T_A:V \to [0,1]$, $I_A:V \to [0,1]$, $F_A:V \to [0,1]$ define respectively a truth membership function,  an indeterminacy membership function, a falsity membership function of the element $x \in X$  to the set $A$ with the condition: $$^-0\leq T_A(x)+I_A(x)+F_A(x)\leq 3^+.$$ The functions $T_A (x)$ , $I_A (x)$ and  $F_A (x)$ are real standard or nonstandard subsets of $] ^{-} 0, 1^{+} [$. 
\end{definition}
Since it is difficult to apply NSs to practical problems, Wang et al. \cite{wa} introduced the concept of SVNS, which is an instance of a NS, and can be used in real scientific and engineering applications.
\begin{definition}\label{ns}\cite{wa}
Let $X$  be a non-empty set.  The neutrosophic set $A$ (NS set) is written as  $A= \{[x : T_A(x), I_A(x), F_A(x)], x \in X\} $ and the functions  $T_A:V \to [0,1]$, $I_A:V \to [0,1]$, $F_A:V \to [0,1]$ denotes the degree of truth membership ,  the degree of indeterminacy membership , the degree of falsity membership of elements of $X$ respectively  and for all $x \in X$ we have $$0\leq T_A(x)+I_A(x)+F_A(x)\leq 3$$ 
\end{definition}

\begin{definition}\label{svng}\cite{bro}
A single valued neutrosophic graph (SVN-graph) with underlying set $V$ is defined to be a pair $G = (A, B)$, where :
\begin{enumerate}
    \item The functions $T_A : V \to [0, 1]$, $I_A: V \to [0, 1]$ and $F_A: V \to [0, 1]$ denote the degree of truth-membership, degree of indeterminacy-membership and falsity-membership of the element $v_i \in V$ respectively, and $0 \leq T_A (v_i) + I_A (v_i)  + F_A(v_i) \leq 3$, for all $v_i \in  V$.
    \item The functions $T_B: E \subseteq  V \times V  \to [0, 1]$, $I_B: E \subseteq  V \times V  \to [0, 1]$ and $F_B: E \subseteq  V \times V  \to [0, 1]$ are defined by 
$$T_B(v_i , v_j) \leq  min [T_A (v_i), T_A (v_j)],$$ 
$$I_B(v_i , v_j) \geq  max [I_A (v_i I_A(v_j)],$$  
$$F_B(v_i,v_j) \geq max[F_A(v_i),F_A (v_j)],$$ denoting the degree of truth-membership, indeterminacy-membership and falsity-membership of the edge $(v_i , v_j) \in E$ respectively, where : 
$$0 \leq  T_B (v_i , v_j) + I_B (v_i , v_j)+ F_B(v_i , v_j) \leq 3,$$ for all $(v_i , v_j) \in E$ ($i, j = 1, 2,\dots , n$).
\end{enumerate}
We call $A$ the single valued neutrosophic vertex set of $V$, and $B$ the single valued neutrosophic edge set of $E$. 

\end{definition}

\begin{definition}\label{fg} \cite{dev} 
Let $G=(V, E)$ be a graph where $V$, $E$ are the set of its vertices and edges respectively.  The fuzzy graph is a pair of functions $F_G = 
(\sigma , \mu )$ where $\sigma$  is a fuzzy subset of a non empty set $V$ and 
$\mu$  is a symmetric fuzzy relation on $\sigma$ . i.e  $\sigma  : V \to  [ 0,1]$
and $\mu: V \times V \to  [0,1]$ such that $\mu(u,v) \leq \mu(u) \wedge \mu(v)$ for 
all $u, v \in V$ where $(u,v)$ denotes the edge between $u$ and $v$ and $\mu(u) \wedge \mu(v)$ denotes the minimum of $\mu(u)$  and $\mu(v)$. $\sigma$ is called the fuzzy vertex set of $V$ and  $\mu$  is called the fuzzy edge set of $E$. 
\end{definition}

\begin{definition}\cite{sma2}
If $X$ is any crisp set, and $A$, $B$ are  two neutrosophic sets. we say that $A=_N B$ if $T_A(x)=T_B(x), I_A(x)=I_B(x), F_A(x)=F_B(x)$ for all $x \in X$ and $A \leq_N B$ if $T_A(x)\leq T_B(x), I_A(x) \leq I_B(x), F_A(x) \leq F_B(x)$ for all $x \in X$
\end{definition}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Some types of $NF$-graphs and operations }
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{definition}\label{g}
Consider the graph $G=(V, E)$ where $V$ and $E$ is the set of its vertex and edges respectively. The neutrosophic fuzzy graph ($NF$-graph) $NF_G=(A,B)$ where $A=[T_A, I_A, F_A]$ and the functions  $T_A:V \to [0,1]$, $I_A:V \to [0,1]$, $F_A:V \to [0,1]$ denotes the degree of truth membership vertex,  the degree of indeterminacy membership vertex, the degree of falsity membership vertex respectively  and for all $v \in V$ we have $$0\leq T_A(v)+I_A(v)+F_A(v)\leq 3$$ and $B=[T_B, I_B, F_B]$ where the functions $T_B:V\times V \to [0,1]$, $I_B:V\times V \to [0,1]$ and $F_B:V\times V \to [0,1]$ defined by  $$T_B(v,w) \leq T_A(v) \times T_A(w),$$
$$I_B(v,w) \leq I_A(v) \times I_A(w),$$
$$F_B(v,w) \leq F_A(v) \times F_A(w),$$ and $$0\leq T_B(v,w)+I_B(v,w)+F_B(v,w)\leq 3,$$
for all $v,w \in V$. $A=[T_A, I_A, F_A]$ and $B=[T_B, I_B, F_B]$  are called the vertex set and the edge set of  $NF_G=(A,B)$ respectively.
\end{definition} 
\begin{definition}\label{c}
A $NF$-graph $NF_G=(A,B)$ is called complete if   $$T_B(v,w) = T_A(v) \times T_A(w),$$
$$I_B(v,w) = I_A(v) \times I_A(w),$$
$$F_B(v,w) = F_A(v) \times F_A(w)$$ for all $v,w \in V$. 
\end{definition} 
\begin{definition}\label{k}
The complement of a $NF$-graph $NF_G=(A,B)$ is the $NF$-graph $\overline{NF_G}=(\overline{A}, \overline{B})$ if\\ $V=\overline{V}$, where $V$, $\overline{V}$ are the vertex set of $G$ and its complement $\overline{G}$ respectively,  \\  $T_A(v) = T_ {\overline{A}} (v)$, $I_A(v) = I_ {\overline{A}}(v)$, $F_A(v) = F_ {\overline{A}} (v)$. 
$$T_{\overline{B}} (v,w) = T_A(v) \times T_A(w) - T_B(v,w),$$
$$I_{\overline{B}} (v,w) = I_A(v) \times I_A(w) - I_B(v,w),$$
 $$F_{\overline{B}} (v,w) = F_A(v) \times F_A(w) - F_B(v,w),$$ for all $v,w \in V$.
\end{definition}
\begin{example}
Consider the following graph $G=(V, E)$ where $V=\{v_1, v_2, v_3\}$ and\\ $E=\{(v_1, v_2), (v_1, v_3), (v_3, v_2) \}$. The $NF$-graph $NF_G=(A,B)$ is defined as in the following diagram
%%%%%%%%%%%%%%%------------------------------------
\end{example}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{figure}[!ht]
    \begin{subfigure}{.5\textwidth}
        \centering
        \begin{tikzpicture}[shorten >=1pt, auto, node distance=6cm, ultra thick]
    \tikzstyle{node_style} = [circle,draw=blue,fill=red!40!,font=\sffamily\Large\bfseries]
    \tikzstyle{edge_style} = [draw=black, line width=2, ultra thick]
    \node[node_style] (v1) at  (0,0) [label=below:$(0.4;0.8; 0.2)$]{$v_1$};
    \node[node_style] (v2) at (4,0) [label=below:$(0.9;0.3; 0.4)$]{$v_2$};
        \node[node_style] (v3) at (2,4) [label=left:$(0.6;0.7; 0.9)$]{$v_3$};
        \draw[edge_style]  (v1) edge  [ above, sloped] node {(0.3,0.2,0.05)} (v2);
       \draw[edge_style]  (v2) edge [ above, sloped]node {(0.5,0.2,0.3)} (v3);
        \draw[edge_style]  (v3) edge[ above, sloped] node  {(0.2,0.5,0.08)} (v1);
   
    \end{tikzpicture}
    \caption{The $NF$-graph $NF_G=(A,B)$}
    
        %\caption*{(a)}
        \label{fig:sfig2}
    \end{subfigure}%
    \begin{subfigure}{.5\textwidth}
        \centering
        \begin{tikzpicture}[shorten >=1pt, auto, node distance=6cm, ultra thick]
    \tikzstyle{node_style} = [circle,draw=blue,fill=red!40!,font=\sffamily\Large\bfseries]
    \tikzstyle{edge_style} = [draw=black, line width=2, ultra thick]
    \node[node_style] (v1) at  (0,0) [label=below:$(0.4;0.8; 0.2)$]{$v_1$};
    \node[node_style] (v2) at (4,0) [label=below:$(0.9;0.3; 0.4)$]{$v_2$};
        \node[node_style] (v3) at (2,4) [label=left:$(0.6;0.7; 0.9)$]{$v_3$};
        \draw[edge_style]  (v1) edge  [ above, sloped] node {(0.06,0.04,0.03)} (v2);
       \draw[edge_style]  (v2) edge [ above, sloped]node {(0.04,0.01,0.06)} (v3);
        \draw[edge_style]  (v3) edge[ above, sloped] node  {(0.04,0.06,0.1)} (v1);
   
    \end{tikzpicture}
    \caption{The $NF$-graph $\overline{NF_G}=(\overline{A},\overline{B})$}
        \label{fig:sfig3}
    \end{subfigure}
    \caption{Two $NF$-graphs one is the complement of the other.}
    \label{fig:figu}
\end{figure}
\begin{remark}
It is obvious that $\overline{\overline{NF_G}}=NF_G$. 
\end{remark}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{definition}\label{plus}
Let $G_1=(V_1,E_1)$, $G_2=(V_2,E_2)$ and $G_2=(V,E)$ be any crisp graphs. The neutrosophic sum of the  NF-graph $NF_{G_1} = (A_1 , B_1 )$ and  $NF_{G_2} = (A_2 , B_2 )$ is the NF-graph $NF_G = (A, B)$ (denoted $NF_G = NF_{G_1} \oplus NF_{G_2}$   if the following conditions are true:
\begin{enumerate}
\item $V_1 = V_2 = V$, where $T_{A_1}(v) = T_{A_2}(v)$, $I_{A_1}(v) = I_{A_2}(v)$, $F_{A_1}(v) = F_{A_2}(v)$ for all $v \in V$,
\item $E=E_1 \cup E_2$ and  for all $(v,w) \in E$.
\end{enumerate}
\[T_{B}(v,w) =
\left\{\begin{array}{r@{\quad:\quad}l} T_{B_1}(v,w) + T_{B_2}(v,w)-T_A(v)\times T_A(w) & T_{B_1}(v,w) + T_{B_2}(v,w) > T_A(v)\times T_A(w); \\ T_{B_1}(v,w) + T_{B_2}(v,w) & otherwise,  \end{array} \right.\]
\[I_{B}(v,w) =
\left\{\begin{array}{r@{\quad:\quad}l} I_{B_1}(v,w) + I_{B_2}(v,w)-I_A(v)\times I_A(w) & I_{B_1}(v,w) + I_{B_2}(v,w) > I_A(v)\times I_A(w); \\ I_{B_1}(v,w) + I_{B_2}(v,w) & otherwise,  \end{array} \right.\]
\[F_{B}(v,w) =
\left\{\begin{array}{r@{\quad:\quad}l} F_{B_1}(v,w) + F_{B_2}(v,w)-F_A(v)\times F_A(w) & F_{B_1}(v,w) + F_{B_2}(v,w) > F_A(v)\times F_A(w); \\ F_{B_1}(v,w) + F_{B_2}(v,w) & otherwise,  \end{array} \right.\]
\end{definition}
 
\begin{example}
Consider the following $NF$-graphs

%========================
\begin{figure}[!ht]
    \begin{subfigure}{.5\textwidth}
        \centering
        \begin{tikzpicture}[shorten >=1pt, auto, node distance=6cm, ultra thick]
    \tikzstyle{node_style} = [circle,draw=blue,fill=red!40!,font=\sffamily\Large\bfseries]
    \tikzstyle{edge_style} = [draw=black, line width=2, ultra thick]
    \node[node_style] (v1) at  (0,0) [label=below:$(0.9;0.4; 0.7)$]{$v_1$};
    \node[node_style] (v2) at (4,0) [label=below:$(0.6;1; 0.3)$]{$v_2$};
        \node[node_style] (v3) at (2,4) [label=left:$(1;0.3; 0.5)$]{$v_3$};
        \draw[edge_style]  (v1) edge  [ above, sloped] node {(0.2,0.2,0.2)} (v2);
       \draw[edge_style]  (v2) edge [ above, sloped]node {(0.5,0.2,0.1)} (v3);
        \draw[edge_style]  (v3) edge[ above, sloped] node  {(0.5,0.1,0.3)} (v1);
   
    \end{tikzpicture}
    \caption{The $NF$-graph $NF_{G_1}=(A_1,B_1)$}
    
        %\caption*{(a)}
        \label{fig:sfig1}
    \end{subfigure}%
    \begin{subfigure}{.5\textwidth}
        \centering
        \begin{tikzpicture}[shorten >=1pt, auto, node distance=6cm, ultra thick]
    \tikzstyle{node_style} = [circle,draw=blue,fill=red!40!,font=\sffamily\Large\bfseries]
    \tikzstyle{edge_style} = [draw=black, line width=2, ultra thick]
    \node[node_style] (v1) at  (0,0) [label=below:$(0.9;0.4; 0.7)$]{$v_1$};
    \node[node_style] (v2) at (4,0) [label=below:$(0.6;1; 0.3)$]{$v_2$};
        \node[node_style] (v3) at (2,4) [label=left:$((1;0.3; 0.5))$]{$v_3$};
        \draw[edge_style]  (v1) edge  [ above, sloped] node {(0.3,0.2,0.2)} (v2);
       \draw[edge_style]  (v2) edge [ above, sloped]node {(0.3,0.2,0.04)} (v3);
        \draw[edge_style]  (v3) edge[ above, sloped] node  {(0.4,0.01,0.2)} (v1);
   
    \end{tikzpicture}
    \caption{The $NF$-graph $NF_{G_2}=(A_2,B_2)$}
        \label{fig:sfig4}
    \end{subfigure}
    \caption{Two $NF$-graphs.}
    \label{fig:fig}
\end{figure}
%======================================================

Then the neotrosophic sum of $NF_{G_1}$ and $NF_{G_2}$ is the $NF$-graph $NF_G =(A,B)$ as follows:
\begin{figure}[!ht]
%\begin{subfigure}[b]{.5\linewidth}
\centering 

\begin{tikzpicture}[shorten >=1pt, auto, node distance=6cm, ultra thick]
    \tikzstyle{node_style} = [circle,draw=blue,fill=red!40!,font=\sffamily\Large\bfseries]
    \tikzstyle{edge_style} = [draw=black, line width=2, ultra thick]
    \node[node_style] (v1) at  (0,0) [label=below:$(0.9;0.4; 0.7)$]{$v_1$};
    \node[node_style] (v2) at (4,0) [label=below:$(0.6;1; 0.3)$]{$v_2$};
        \node[node_style] (v3) at (2,4) [label=left:$(1;0.3; 0.5)$]{$v_3$};
        \draw[edge_style]  (v1) edge  [ above, sloped] node {(0.5,0.4,0.19)} (v2);
       \draw[edge_style]  (v2) edge [ above, sloped]node {(0.2,0.1,0.14)} (v3);
        \draw[edge_style]  (v3) edge[ above, sloped] node  {(0.8,0.11,0.15)} (v1);
   
    \end{tikzpicture}
    \caption{The $NF$-graph $NF_G = NF_{G_1} \oplus NF_{G_2} $}
    
\end{figure}

\end{example}
\begin{proposition}
Let $G_1=(V, E_1)$, $G_2=(V, E_2)$ be two crisp graphs with the corresponding NF-graphs $NF_{G{_1}}=(A_1, B_1)$, $NF_{G_2}=(A_2,B_2)$.  Then their neutrosophic sum is complete if and only if one of them is  neutrosophic complement of the other. 
 \end{proposition}
 \begin{proof}
  Let $NF_{G{_1}}=(A_1, B_1)$ and $NF_{G_2}=(A_2,B_2)$ be neutrosophic complement of each other and  suppose that $NF_H=(A,B)$ is their neutrosophic sum  that is $NF_H = NF_{G_{1}} \oplus NF_{G_2}$ . Hence, by Definition \ref{c}, we have $$T_{B_1}(v,w)=T_A(v) \times T_A(w) - T_{B_2}(v,w),$$ $$I_{B_1}(v,w)=I_A(v) \times I_A(w) - I_{B_2}(v,w),$$ $$F_{B_1}(v,w)=F_A(v) \times F_A(w) - F_{B_2}(v,w),$$ for all $(v,w)\in E$.
  
From  Definition \ref{plus}, we have
 
$$T_{B}(v,w) = T_{B_1}(v,w) + T_{B_2}(v,w),$$  $$I_{B}(v,w) = I_{B_1}(v,w) + I_{B_2}(v,w),$$ $$F_{B}(v,w) =F_{B_1}(v,w) + F_{B_2}(v,w),$$ for all $(v,w) \in E$. Substituting we obtain $$T_{B}(v,w) = T_A(v) \times T_A(w),$$  $$I_{B}(v,w) = I_A(v) \times I_A(w),$$ $$F_{B}(v,w) =F_A(v) \times F_A(w),$$ for all $(v,w) \in E$. This implies that $NF_H=(A,B)$ is complete.\\ Conversely, by reversing the steps, we get the result.  
 \end{proof}
\begin{proposition}
Let $\overline{NF_G}=(\overline{A}, \overline{B})$ be the complement of the $NF$-graph $NF_G=(A,B)$.  Then  $NF_G=(A,B)$ is a complete NF-graph if and only if $$T_{\overline{B}} (v,w) = I_{\overline{B}} (v,w)= F_{\overline{B}} (v,w) =0 \quad \forall v,w \in V.$$
 \end{proposition}
\begin{proof} Suppose that $NF_G=(A,B)$ is a complete NF-graph, then by Definition \ref{c}, we have $$T_B(v,w) = T_A(v) \times T_A(w),$$
$$I_B(v,w) = I_A(v) \times I_A(w),$$
$$F_B(v,w) = F_A(v) \times F_A(w)$$ for all $v,w \in V$. From Definition \ref{k}, $$T_{\overline{B}} (v,w) = T_A(v) \times T_A(w) - T_B(v,w),$$
$$I_{\overline{B}} (v,w) = I_A(v) \times I_A(w) - I_B(v,w),$$
 $$F_{\overline{B}} (v,w) = F_A(v) \times F_A(w) - F_B(v,w),$$ for all $v,w \in V$. Hence, $$T_{\overline{B}} (v,w) = I_{\overline{B}} (v,w)= F_{\overline{B}} (v,w) =0 \quad \forall v,w \in V.$$ conversely, if $$T_{\overline{B}} (v,w) = I_{\overline{B}} (v,w)= F_{\overline{B}} (v,w) =0 \quad \forall v,w \in V.$$ Then we obtain that  $$0 = T_A(v) \times T_A(w) - T_B(v,w),$$
$$0 = I_A(v) \times I_A(w) - I_B(v,w),$$
 $$0 = F_A(v) \times F_A(w) - F_B(v,w),$$ for all $v,w \in V$. Hence, $$T_B(v,w) = T_A(v) \times T_A(w),$$
$$I_B(v,w) = I_A(v) \times I_A(w),$$
$$F_B(v,w) = F_A(v) \times F_A(w)$$ for all $v,w \in V$. Implies $NF_G=(A,B)$ is a complete NF-graph.
 \end{proof}
\begin{definition}
Let $H=(V_1,E_1)$ and $G=(V_2,E_2)$ be any two graphs. A NF-graph $NF_H = (A_1 , B_1 )$  is partial NF-subgraph of the NF-graph $NF_G = (A_2, B_2)$  if the following conditions are true:
\begin{enumerate}
\item $V_1 \subseteq V_2$, where $T_{A_1}(v) \leq T_{A_2}(v)$, $I_{A_1}(v) \leq I_{A_2}(v)$, $F_{A_1}(v) \leq F_{A_2}(v)$ for all $v \in V_1$,
\item $E_1 \subseteq E_2$, where $T_{B_1}(v,w) \leq T_{B_2}(v,w)$, $I_{B_1}(v,w) \leq I_{B_2}(v,w)$, $F_{B_1}(v,w) \leq F_{B_2}(v,w)$ for all $(v,w) \in E_1$.
\end{enumerate}

\end{definition}
\begin{definition}\label{sg}
Let $H=(V_1,E_1)$ and $G=(V_2,E_2)$ be any two graphs. A NF-graph $NF_H = (A_1 , B_1 )$  is called a NF-subgraph of the NF-graph $NF_G = (A_2, B_2)$  if the following conditions are true:
\begin{enumerate}
\item $V_1 \subseteq V_2$, where $T_{A_1}(v) = T_{A_2}(v)$, $I_{A_1}(v) = I_{A_2}(v)$, $F_{A_1}(v) = F_{A_2}(v)$ for all $v \in V_1$,
\item $E_1 \subseteq E_2$, where $T_{B_1}(v,w) = T_{B_2}(v,w)$, $I_{B_1}(v,w) = I_{B_2}(v,w)$, $F_{B_1}(v,w) = F_{B_2}(v,w)$ for all $(v,w) \in E_1$.
\end{enumerate}
\end{definition}

\begin{definition}\label{ssg}
Let $H=(V,E_1)$ and $G=(V,E_2)$ be any two graphs. A NF-graph $NF_H = (A_1 , B_1 )$  is called a spanning NF-subgraph of the NF-graph $NF_G = (A_2, B_2)$  if the following conditions are true:
\begin{enumerate}
\item  $T_{A_1}(v) = T_{A_2}(v)$, $I_{A_1}(v) = I_{A_2}(v)$, $F_{A_1}(v) = F_{A_2}(v)$ for all $v \in V$,
\item $E_1 \subseteq E_2$, where $T_{B_1}(v,w) = T_{B_2}(v,w)$, $I_{B_1}(v,w) = I_{B_2}(v,w)$, $F_{B_1}(v,w) = F_{B_2}(v,w)$ for all $ (v,w) \in E_1$.
\end{enumerate}
\end{definition}
\begin{definition}
Let $H_1=(V,E_1)$ and $H_2=(V,E_2)$ be two crisp graphs and $NF_{H_1} = (A_1 , B_1 )$, $NF_{H_2} = (A_2 , B_2 )$ are two spanning NF-subgraphs  of the NF-graph $NF_G = (A, B)$, we call that $NF_{H_1} \sqcap NF_{H_2} = \Phi$ if $E_1 \cap E_2 = \phi$.\\ Moreover, if $G=(V,E)$, $H_1(V, E_1)$, $H_2(V, E_2)$, $\dots$ are crisp graphs such that  $NF_{H_1} = (A_1 , B_1 )$, $NF_{H_2} = (A_2 , B_2 )$, $\dots$ are  spanning NF-subgraphs  of the NF-graph $NF_G = (A, B)$. If $E= \bigcup_{i=1}^\infty E_i$, then we say that $$NF_G = \bigsqcup_{i=1}^\infty NF_{H_i}$$.
\end{definition}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{proposition}
Let $G=(V, E)$ be any crisp graph. Then the $NF$-graph $NF_G=(A,B)$ is a complete $NF$-graph if and only if $NF_H \oplus NF_H = NF_H$ for every spanning NF-subgraph $NF_H$ of the $NF$-graph $NF_G$.   
\end{proposition}
\begin{proof}
Suppose that $NF_G=(A,B)$ is complete and $NF_H$ is any spanning subgraph of it. Since $NF_G=(A,B)$ is complete, so by Definition \ref{c}, we have $$T_B(v,w) = T_A(v) \times T_A(w),$$
$$I_B(v,w) = I_A(v) \times I_A(w),$$
$$F_B(v,w) = F_A(v) \times F_A(w)$$ for all $v,w \in V$. Since $NF_H$ is a spanning subgraph of $NF_G$, so we obtain that $$T_{B_1}(v,w) = T_A(v) \times T_A(w),$$
$$I_{B_1}(v,w) = I_A(v) \times I_A(w),$$
$$F_{B_1}(v,w) = F_A(v) \times F_A(w)$$ for all $v,w \in V$. Hence, by Definition \ref{plus}, we deduce that the truth, intermidacy  and fulsity membership of $NF_H \oplus NF_H$ is    $$T_{B_1}(v,w) = 2 T_A(v) \times T_A(w) - T_A(v) \times T_A(w)=T_A(v) \times T_A(w),$$
$$I_{B_1}(v,w) = 2 I_A(v) \times I_A(w) - I_A(v) \times I_A(w)=I_A(v) \times I_A(w),$$
$$F_{B_1}(v,w) = 2 F_A(v) \times F_A(w) - F_A(v) \times F_A(w)=F_A(v) \times F_A(w)$$ for all $v,w \in V$. Hence, $NF_H \oplus NF_H = NF_H$.\\ Conversely, Since $NF_H \oplus NF_H = NF_H$ for every spanning NF-subgraph $NF_H$ of the $NF$-graph $NF_G$ and $NF_G$ is a spanning  NF-subgraph of itself, so by hypothesis  $NF_G \oplus NF_G = NF_G$. Therefore, by Definition \ref{plus}, we get $$T_B(v,w) + T_B(v,w)- T_A(v) \times T_A(w)  = T_B(v,w),$$
$$I_B(v,w) + I_B(v,w) - I_A(v) \times I_A(w) = I_B(v,w),$$
$$F_B(v,w) + F_B(v,w) - F_A(v) \times F_A(w) = F_B(v,w)$$ for all $v,w \in V$. This implies that $$T_B(v,w) = T_A(v) \times T_A(w),$$
$$I_B(v,w) = I_A(v) \times I_A(w),$$
$$F_B(v,w) = F_A(v) \times F_A(w)$$ for all $v,w \in V$. Therefore, $NF_G$  is complete. 
\end{proof}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Degrees and strong degrees in $NF$-graphs}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{definition}\label{d}
Consider the graph $G=(V, E)$ where $V$ and $E$ is the set of its vertex and edges respectively.  $NF_G=(A,B)$ be the $NF$-graph. Then the degree of the vertex $v \in V$  is denoted by $d(v)$  and defined as $d(v)=[d_T(v), d_I(v), d_F(v)]$ (or $d_G(v)=[d_{T_G}(v), d_{I_G}(v), d_{F_G}(v)]$)  where $$d_T(v)= \sum_{v\neq w} T_B(v, w), \qquad d_I(v)= \sum_{v\neq w} I_B(v, w), \qquad d_F(v)= \sum_{v\neq w} F_B(v, w)$$ denotes the degree of truth membership vertex,  the degree of indeterminacy membership vertex, the degree of falsity membership vertex respectively.   If $T_B(v, w)=I_B(v, w)=F_B(v, w)=0$ for $v,w \in V$, then we say that they are not adjacent.
\end{definition}
\begin{definition}\label{t}
Consider the graph $G=(V, E)$ where $V$ and $E$ is the set of its vertex and edges respectively.  $NF_G=(A,B)$ be the $NF$-graph. Then the total degree of the vertex $v \in V$  is denoted by $td(v)$ and defined as $td(v)=[td_T(v), td_I(v), td_F(v)]$ where $$td_T(v)= \sum_{v\neq w} T_B(v, w) + T_A(v), \qquad td_I(v)= \sum_{v\neq w} I_B(v, w)+ I_A(v), \qquad td_F(v)= \sum_{v\neq w} F_B(v, w) + F_A(v).$$
\end{definition}
\begin{definition}\label{n}
Consider the graph $G=(V, E)$ where $V$ and $E$ are the set of its vertex and edges respectively. The neighborhood of a vertex $v \in V$ is denoted by $\mathcal{N}(v)$ and is defined as  $\mathcal{N}(v) = \{ w \in V : (v,w) \in E\}$.
\end{definition}
\begin{example}
Consider the following graph $G=(V, E)$ where $V=\{v_1, v_2, v_3, v_4\}$ and\\ $E=\{(v_1, v_2), (v_1, v_3), (v_2, v_4), (v_3, v_4), (v_3, v_2) \}$. The $NF$-graph $NF_G=(A,B)$ is defined as in the following diagram
\end{example}


%%%%%%%%%%%%%%%------------------------------------
\begin{figure}[!ht]
\centering 
\begin{tikzpicture}[shorten >=1pt, auto, node distance=6cm, ultra thick]
    \tikzstyle{node_style} = [circle,draw=blue,fill=red!40!,font=\sffamily\Large\bfseries]
    \tikzstyle{edge_style} = [draw=black, line width=2, ultra thick]
    \node[node_style] (v1) at  (-2,2) [label=left:$(0.1;0.3; 0.7)$]{$v_1$};
    \node[node_style] (v2) at (2,2) [label=right:$(0.5;0.8; 0.3)$]{$v_2$};
    %\node[node_style] (v3) at (4,0) [label=right:$(0.9;0.6; 0.4)$]{$v_6$};
    \node[node_style] (v4) at (2,-2) [label=right:$(0.9;0.6; 0.4)$]{$v_4$};
    \node[node_style] (v3) at (-2,-2) [label=left:$(0.6;0.4; 0.2)$]{$v_3$};
        \draw[edge_style]  (v1) edge  node {(0.05,0.2,0.2)} (v2);
       \draw[edge_style]  (v2) edge [ above, sloped]node {(0.2,0.3,0.05)} (v3);
    \draw[edge_style]  (v3) edge[ below, sloped] node {(0.5,0.2,0.05)} (v4);
    \draw[edge_style]  (v3) edge[ above, sloped] node  {(0.05,0.1,0.1)} (v1);
        \draw[edge_style]  (v4) edge[ below, sloped] node {(0.4,0.4,0.1)}(v2)  ;
    \end{tikzpicture}
    \caption{$NF$-graph $NF_G=(A,B)$}
\end{figure}
By simple calculation, we get\\
$d(v_1)=[d_T(v_1), d_I(v_1), d_F(v_1)] = [0.1, 0.3, 0.3]$, $d(v_2)=[d_T(v_2), d_I(v_2), d_F(v_2)] = [0.65, 0.9, 0.35]$ \\
$d(v_3)=[d_T(v_3), d_I(v_3), d_F(v_3)] = [0.75, 0.6, 0.2]$, $d(v_4)=[d_T(v_4), d_I(v_4), d_F(v_4)] = [0.9, 0.6, 0.15]$.\\
$td(v_1)= [0.2, 0.6, 1]$, $td(v_2)= [1.15, 1.7, 0.65]$ \\
$td(v_3)= [1.35, 1, 0.4]$, $td(v_4)= [1.8, 1.2, 0.55]$.
\begin{proposition}\label{p1}
Let $G=(V, E)$ where $V$  is its vertex set  and $E$ its edge set.  $NF_G=(A,B)$ be the $NF$-graph. Then the degree of truth membership vertex,  the degree of indeterminacy membership vertex, the degree of falsity membership vertex respectively is of the vertex $v \in V$  is given by   $$d_T(v) \leq T_A(v)( \sum_{w \in \mathcal{N}(v)} T_A(w)),$$  $$ d_I(v)\leq I_A(v)( \sum_{w \in \mathcal{N}(v)} I_A(w)),$$ $$d_F(v) \leq F_A(v)( \sum_{w \in \mathcal{N}(v)} F_A(w))$$ and  $$td_T(v) \leq T_A(v)(\sum_{w \in \mathcal{N}(v)} T_A(w) +1),$$  $$ td_I(v)\leq I_A(v)(\sum_{w \in \mathcal{N}(v)} I_A(w) +1),$$ $$td_F(v) \leq F_A(v)(\sum_{w \in \mathcal{N}(v)} F_A(w) +1)$$
\end{proposition}
\begin{proof}
From Definition \ref{d}, we have $$d_T(v)= \sum_{v\neq w} T_B(v, w), \qquad d_I(v)= \sum_{v\neq w} I_B(v, w), \qquad d_F(v)= \sum_{v\neq w} F_B(v, w)$$ and from Definition \ref{g}, we have $$T_B(v,w) \leq T_A(v) \times T_A(w),$$
$$I_B(v,w) \leq I_A(v) \times I_A(w),$$
$$F_B(v,w) \leq F_A(v) \times F_A(w).$$
Since $T_B(v,w) =I_B(v,w) =F_B(v,w) =0 \quad \forall (v,w) \notin E$ Therefore,
%\begin{equation}\label{1}
$$d_T(v) \leq \sum_{w \in \mathcal{N}(v)} T_A(v)( T_A(w)) = T_A(v) (\sum_{w \in \mathcal{N}(v)}  T_A(w)),$$  $$ d_I(v)\leq  \sum_{w \in \mathcal{N}(v)} I_A(v)(  I_A(w)) = I_A(v)( \sum_{w \in \mathcal{N}(v)}   I_A(w)),$$ $$d_F(v) \leq \sum_{w \in \mathcal{N}(v)} F_A(v)(  F_A(w)) = F_A(v)( \sum_{w \in \mathcal{N}(v)}  F_A(w)).$$
%\end{equation} 
  
From Definition \ref{t}, and above equations, we obtain that 
%\begin{equation}\label{2}
$$td_T(v) \leq \sum_{w \in \mathcal{N}(v)} T_A(v)( T_A(w)) + T_A(v)  = T_A(v) (\sum_{w \in \mathcal{N}(v)}  T_A(w)+1),$$  $$ td_I(v)\leq  \sum_{w \in \mathcal{N}(v)} I_A(v)(  I_A(w)) + I_A(v) = I_A(v)( \sum_{w \in \mathcal{N}(v)}   I_A(w)+1),$$ $$td_F(v) \leq \sum_{w \in \mathcal{N}(v)} F_A(v)(  F_A(w)) + F_A(v) = F_A(v)( \sum_{w \in \mathcal{N}(v)}  F_A(w)+1).$$%\end{equation}
\end{proof}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{proposition}
Let $NF_G=(A,B)$ be a complete $NF$-graph. Then  $$d_T(v) = T_A(v)( \sum_{w \in \mathcal{N}(v)} T_A(w)),$$  $$ d_I(v)= I_A(v)( \sum_{w \in \mathcal{N}(v)} I_A(w)),$$ $$d_F(v) = F_A(v)( \sum_{w \in \mathcal{N}(v)} F_A(w))$$ and  $$td_T(v) = T_A(v)(\sum_{w \in \mathcal{N}(v)} T_A(w) +1),$$  $$ td_I(v)= I_A(v)(\sum_{w \in \mathcal{N}(v)} I_A(w) +1),$$ $$td_F(v) = F_A(v)(\sum_{w \in \mathcal{N}(v)} F_A(w) +1)$$
\end{proposition}
\begin{proof}
Follows from Definition \ref{c} and Proposition \ref{p1}.
\end{proof}
\begin{proposition}
\begin{enumerate}
    \item Let $NF_H=(C, D)$  be any partial subgraph of the $NF$-graph $NF_G=(A,B)$ . Then $$d_{T_H}(v) \leq d_{T_G}(v),\quad d_{I_H}(v) \leq d_{I_G}(v), \quad d_{F_H}(v) \leq d_{F_G}(v),$$  and  $$td_{T_H}(v) \leq td_{T_G}(v), \quad td_{I_H}(v) \leq td_{I_G}(v),\quad td_{F_H}(v) \leq td_{F_G}(v),$$ for all $v \in V(H)$.
    \item Let $G=(V,E)$ be any graph and let $\{v_1, v_2, \dots \}$ be the set of isolated vertices of $G$. If $H=(V_1, E_1)$ is a subgraph of $G$ such that $V_1 = V \setminus \{v_i : i \in N\}$. Then   $$d_{T_H}(v) = d_{T_G}(v),\quad d_{I_H}(v) = d_{I_G}(v), \quad d_{F_H}(v) = d_{F_G}(v),$$  and  $$td_{T_H}(v) = td_{T_G}(v), \quad td_{I_H}(v) = td_{I_G}(v),\quad td_{F_H}(v) = td_{F_G}(v),$$ for all $v \in V(H)$.
\end{enumerate}
  
\end{proposition}
\begin{proof}
(1) Follows from the definition.\\
(2) Since $\{v_1, v_2, \dots \}$ is the set of isolated vertices of $G$. Hence $T_B(v,v_i) =I_B(v,v_i) =F_B(v,v_i) =0 \quad \forall i=1,2, \dots$. Therefore $$d_{T_H}(v) = d_{T_G}(v),\quad d_{I_H}(v) = d_{I_G}(v), \quad d_{F_H}(v) = d_{F_G}(v),$$  and  $$td_{T_H}(v) = td_{T_G}(v), \quad td_{I_H}(v) = td_{I_G}(v),\quad td_{F_H}(v) = td_{F_G}(v),$$ for all $v \in V(H)$.
\end{proof}
\begin{proposition}
Let $G=(V, E)$ where $V$  is its vertex set  and $E$ its edge set.  $NF_G=(A,B)$ be the $NF$-graph. Then 
%\begin{equation}\label{3}
$$\sum_{v \in V} d_T(v)= 2 \sum_{v\neq w} T_B(v, w),$$  $$\sum_{v \in V} d_I(v)= 2 \sum_{v\neq w} I_B(v, w),$$ $$\sum_{v \in V} d_F(v)=2 \sum_{v\neq w} F_B(v, w).$$
%\end{equation}
\end{proposition}
\begin{proof} Since  $T_B(v, w)=T_B(w, v)$, $I_B(v, w)=I_B(w, v)$, $F_B(v, w)=F_B(w, v)$ for every $v,w \in V$, so in $\sum_{v \in V} d_T(v)$, $\sum_{v \in V} d_I(v)$ and $\sum_{v \in V} d_F(v)$, we get  $T_B(v, w) + T_B(w, v)$, $I_B(v, w) + I_B(w, v)$, $F_B(v, w) + F_B(w, v)$. Hence, the result.
\end{proof}
\begin{definition}
Let $G=(V,E)$ be any graph, an edge $e = (v, w)\in E$, then the corresponding $NF$-edge of the NF-graph  $NF_G = (A,B)$ is called a strong edge (denoted by $E_s$), if $$T_B(v,w) = T_A(v) \times T_A(w),$$
$$I_B(v,w) = I_A(v) \times I_A(w),$$
$$F_B(v,w) = F_A(v) \times F_A(w).$$
 In this case, the vertex $v$ is called a strong neighbor of $w$ and conversely.\\
$\mathcal{SN}(v) = \{w \in V : (u,w) \in E_s \}$ is called the strong neighborhood of $v$. and the set $\mathcal{SN}[v] =\{v\} \cup \mathcal{SN}(v)$ is called the closed strong neighborhood of $v$.
\end{definition}

\begin{example}\label{e3}
Consider the following graph $G=(V, E)$ where $V=\{v_1, v_2, v_3, v_4, v_5\}$ and\\ $E=\{(v_1, v_2), (v_1, v_3), (v_2, v_4), (v_3, v_4), (v_3, v_2) \}$. The $NF$-graph $NF_G=(A,B)$ is defined as in the following diagram:

\begin{figure}[hbt!]
\centering 
\begin{tikzpicture}[shorten >=1pt, auto, node distance=6cm, ultra thick]
    \tikzstyle{node_style} = [circle,draw=blue,fill=red!40!,font=\sffamily\Large\bfseries]
    \tikzstyle{edge_style} = [draw=black, line width=2, ultra thick]
    \node[node_style] (v1) at  (-2,3) [label=left:$(0.6;0.4; 0.8)$]{$v_1$};
    \node[node_style] (v2) at (2,3) [label=right:$(0.5;1; 0.2)$]{$v_2$};
    \node[node_style] (v5) at (6,0) [label=right:$(0.9;0.6; 0.4)$]{$v_5$};
    \node[node_style] (v4) at (2,-3) [label=right:$(0.6;0.2; 0.7)$]{$v_4$};
    \node[node_style] (v3) at (-2,-2) [label=left:$(0.8;0.4; 0.5)$]{$v_3$};
        \draw[edge_style]  (v1) edge  node {(0.3,0.4,0.16)} (v2);
       \draw[edge_style]  (v1) edge [ above, sloped]node {(0.36,0.08,0.56)} (v4);
    \draw[edge_style]  (v3) edge[ below, sloped] node {(0.4,0.02,0.3)} (v4);
    \draw[edge_style]  (v3) edge[ above, sloped] node  {(0.4,0.1,0.4)} (v1);
        \draw[edge_style]  (v4) edge[ below, sloped] node {\qquad(0.3,0.2,0.14)}(v2)  ;
        \draw[edge_style]  (v2) edge[ below, sloped] node {(0.45,0.6,0.08)} (v5);
    \draw[edge_style]  (v3) edge[ below, sloped] node  {(0.7,0.2,0.2)} (v5);
        \draw[edge_style]  (v4) edge[ below, sloped] node {(0.54,0.12,0.28)}(v5)  ;
    \end{tikzpicture}
    \caption{$NF$-graph $NF_G=(A,B)$}
\end{figure}
\end{example}
It is easy to see that $\mathcal{SN}(v_1) = \{v_2, v_4 \}$, $\mathcal{SN}(v_2) = \{v_1, v_4, v_5 \}$, $\mathcal{SN}(v_3) = \phi$, $\mathcal{SN}(v_4) = \{v_1, v_2, v_5 \}$ and $\mathcal{SN}(v_5) = \{v_2, v_4 \}$.

\begin{definition}\label{sd}
Consider the graph $G=(V, E)$ where $V$ and $E$ is the set of its vertex and edges respectively.  $NF_G=(A,B)$ be the $NF$-graph. Then the strong degree of the vertex $v \in V$  is denoted by $sd(v)$ and defined as $sd(v)=[sd_T(v), sd_I(v), sd_F(v)]$ where $$sd_T(v)= \sum_{ w \in \mathcal{SN}(v) } T_B(v, w), \qquad sd_I(v)= \sum_{w \in \mathcal{SN}(v)} I_B(v, w), \qquad sd_F(v)= \sum_{w \in \mathcal{SN}(v)} F_B(v, w)$$ denotes the strong degree of truth, indeterminacy and falsity  membership vertex $v$ respectively.   If $sT_B(v, w)=sI_B(v, w)=sF_B(v, w)=0$ for some $v,w \in V$, then we say that $w \in \mathcal{SN}(v)$.
\end{definition}
\begin{definition}\label{st}
Consider the graph $G=(V, E)$ where $V$ and $E$ is the set of its vertex and edges respectively.  $NF_G=(A,B)$ be the $NF$-graph. Then the total strong degree of the vertex $v \in V$  is denoted by $tsd(v)$ and defined as $tsd(v)=[tsd_T(v), tsd_I(v), tsd_F(v)]$ where $$tsd_T(v)=\sum_{w \in \mathcal{SN}(v)} T_B(v, w) + T_A(v),$$ $$tsd_I(v)=\sum_{w \in \mathcal{SN}(v)} I_B(v, w)+ I_A(v),$$ $$tsd_F(v)=\sum_{w \in \mathcal{SN}(v)} F_B(v, w) + F_A(v).$$
\end{definition}
From Example \ref{e3}, we conclude that\\ $sd(v_1)=[0.66, 0.48, 0.72]$, $sd(v_2)=[1.05, 1.2, 0.38]$, $sd(v_3)=[0, 0, 0]$, $sd(v_4)=[1.2, 0.3, 0.98]$ and $sd(v_5)=[0.99, 0.72, 0.36]$.\\ $tsd(v_1)=[1.26, 0.88, 1.52]$, $tsd(v_2)=[1.55, 2.2, 0.58]$, $tsd(v_3)=[0.8, 0.4, 0.5]$, $tsd(v_4)=[1.8, 0.5, 1.68]$ and $sd(v_5)=[1.89, 1.32, 0.76]$.
\begin{proposition}
Let $G=(V,E)$ be any graph and $NF_G(A,B)$ is its corresponding NF-graph, then the following are true:
\begin{enumerate}
\item If $v \in V$ is an isolated vertex, then $d(v)=[0,0,0]$ and $td(v)=[T_A(v),I_A(v),F_A(v)]$.
\item If $\mathcal{SN}(v) = \phi$, then $sd(v)=[0,0,0]$ and $tsd(v)=[T_A(v),I_A(v),F_A(v)]$.
\item If $v \in V$ is an isolated vertex, then $d(v)=sd(v)$ and $td(v)=tsd(v)$.
\item $sd(v) \leq d(v)$ and $tsd(v) \leq td(v)$ for all $v \in V$.
\item The NF-graph $NF_G(A,B)$ is complete if and only if $sd(v) = d(v)$ and $tsd(v) = td(v)$ for all $v \in V$.
\end{enumerate}
\end{proposition}
\begin{proof}

 \item  [(1), (2)] The proof of (1) and (2) follow from their definitions.

\item[(3)]Follows from (1) and (2).
\item[(4)]Since $sd(v)=[sd_T(v), sd_I(v), sd_F(v)]$ and $sd_T(v), sd_I(v), sd_F(v)$ is a sum taken from only strong edges which are adjacent to $v$, so this sum is less than or equal to the sum taken over all edges adjacent to $v$. Therefore, $sd(v) \leq d(v)$ and if we add $T_A(v),I_A(v),F_A(v)$ to both sides of the inequality respectively, then we get $tsd(v) \leq td(v)$ for all $v \in V$.
\item[(5)]Suppose that $NF_G(A,B)$ is a complete graph, then by definition every edge is strong and hence $sd(v) = d(v)$ and $tsd(v) = td(v)$ for all $v \in V$. Now if   $sd(v) = d(v)$ for all $v \in V$ implies that $$\sum_{ w \in \mathcal{N}(v) } T_B(v, w)=d_T(v)=sd_T(v)= \sum_{ w \in \mathcal{SN}(v) } T_B(v, w), $$ and this will be true only when $\mathcal{SN}(v) = \mathcal{N}(v)$ for all $v \in V$. Hence, every edge is strong means that $$T_B(v,w) = T_A(v) \times T_A(w),$$
$$I_B(v,w) = I_A(v) \times I_A(w),$$
$$F_B(v,w) = F_A(v) \times F_A(w)$$ for all $(v,w) \in E$. Therefore, $NF_G(A,B)$ is a complete graph. 
\end{proof} 

\begin{proposition}
if $G=(V,E)$, $H_1(V, E_1)$, $H_2(V, E_2)$, $\dots$ are crisp graphs such that  $NF_{H_1} = (A_1 , B_1 )$, $NF_{H_2} = (A_2 , B_2 )$, $\dots$ are  spanning NF-subgraphs  of the NF-graph $NF_G = (A, B)$ with  $E= \bigcup_{i=1}^\infty E_i$. Then $$NF_G = \bigsqcup_{i=1}^\infty NF_{H_i},$$ and $$d_T(v) = \sum_{i=1}^\infty  d^i_T(v) -  \sum_{(u,v) \in E_i \cap E_j} T_{B}(u,v),$$ $$d_I(v) = \sum_{i=1}^\infty  d^i_I(v) -  \sum_{(u,v) \in E_i \cap E_j} I_{B}(u,v),$$ $$d_F(v) = \sum_{i=1}^\infty  d^i_F(v) -  \sum_{(u,v) \in E_i \cap E_j} F_{B}(u,v),$$ where $d^i_T(v)$, $d^i_I(v)$ and $d^i_F(v)$ denotes the degree of truth membership vertex,  the degree of indeterminacy membership vertex, the degree of falsity membership vertex respectively in the neuotrosophic soft graph $NF_{H_i}$. Moreover, if $E_i$ is pairwise disjoint for all $i=1,2, \dots$, then
$$d_T(v) = \sum_{i=1}^\infty  d^i_T(v) ,$$ $$d_I(v) = \sum_{i=1}^\infty  d^i_I(v) ,$$ $$d_F(v) = \sum_{i=1}^\infty  d^i_F(v) .$$  
\end{proposition}
\begin{proof}
Since for each edge $e=(u,v)\in E_i \cap E_j$, we have $d^i_T(v)$ and $d^j_T(v)$  contain the term $T_{B_i}(u,v)= T_{B}(u,v)$, $T_{B_i}(u,v) =T_{B_i}(u,v)$ respectively. Hence, we conclude that $d_T(v) = d^i_T(v) + d^j_T(v) - T_{B}(u,v)$ whenever $E_i \cap E_j$ contains  the only edge $e = (u,v)$. Hence if $E_i \cap E_j$ more than one edge, so we subtract them from the sum of the degree of truth membership vertex $v \in V$. Similarly, the degree of indeterminacy membership vertex, the degree of falsity membership  are obtained.\\ If $E_i \cap E_j = \phi$ for all $1 \leq i,j $, so if $T_{B}(u,v) \neq 0$ in $NF_{H_i}$, then  $T_{B}(u,v) = 0$ in $NF_{H_j}$ this means that $T_{B}(u,v)$ will occur only one time. Hence, the result.
\end{proof}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
  \section{Conclusions} 
  The neutrosophic strong degree and total strong degree of the vertex of graphs were introduced with interesting properties on them. Furthermore, this paper discussed the concepts of strong edge, neutrosophic sum in NF – graph with suitable examples.  

  \noindent{\bf Conflicts of Interest:} The authors declare no conflict of interest.

%===============================================





\bigskip
\bigskip
\nt {\footnotesize {\bf Received: date / Accepted: date }





\end{document}
